

GOPEN ACCESS

Citation: Gakuya F, Akoko J, Wambua L, Nyamota R, Ronoh B, Lekolool I, et al. (2022) Evidence of co-exposure with *Brucella* spp, *Coxiella burnetii*, and Rift Valley fever virus among various species of wildlife in Kenya. PLoS Negl Trop Dis 16(8): e0010596. https://doi.org/10.1371/journal. pntd.0010596

Editor: Ana LTO Nascimento, Instituto Butantan, BRAZIL

Received: December 10, 2021

Accepted: June 20, 2022

Published: August 8, 2022

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This study was sponsored by the United States Department of the Defense, Defense Threat Reduction Agency (HDTRA1-19-1-0031). The content of this paper does not necessarily reflect the position or the policy of the federal government, and no official endorsement should be inferred. The funders had no role in study RESEARCH ARTICLE

Evidence of co-exposure with *Brucella* spp, *Coxiella burnetii*, and Rift Valley fever virus among various species of wildlife in Kenya

Francis Gakuya¹[®], James Akoko^{2,3}[®]*, Lillian Wambua², Richard Nyamota², Bernard Ronoh⁴, Isaac Lekolool⁴, Athman Mwatondo^{2,5,6}, Mathew Muturi^{2,5,7}, Collins Ouma³, Daniel Nthiwa⁸, Earl Middlebrook⁹, Jeanne Fair⁹, John Gachohi^{10,11}, Kariuki Njenga¹⁰, Bernard Bett²

1 Wildlife Research and Training Institute, Naivasha, Kenya, 2 International Livestock Research Institute, Nairobi, Kenya, 3 Department of Biomedical Science, Maseno University, Kisumu, Kenya, 4 Kenya Wildlife Service, Nairobi, Kenya, 5 Zoonotic Disease Unit, Nairobi, Kenya, 6 Department of Medical Microbiology and Immunology, Faculty of Health, University of Nairobi, Kenya, 7 Faculty of Veterinary Medicine, Freie Universität Berlin, Germany, 8 Department of Biological Sciences, University of Embu, Embu, Kenya, 9 Los Alamos National Laboratory, New Mexico, United States of America, 10 Washington State University, Global Health Programme, Nairobi, Kenya, 11 School of Public Health, Jomo Kenyatta University of Agriculture and Technology, Nairobi, Kenya

• These authors contributed equally to this work.

* jamesakoko@yahoo.com

Abstract

Background

Co-infection, especially with pathogens of dissimilar genetic makeup, may result in a more devastating impact on the host. Investigations on co-infection with neglected zoonotic pathogens in wildlife are necessary to inform appropriate prevention and control strategies to reduce disease burden in wildlife and the potential transmission of these pathogens between wildlife, livestock and humans. This study assessed co-exposure of various Kenyan wildflife species with *Brucella spp, Coxiella burnetii* and Rift Valley fever virus (RVFV).

Methodology

A total of 363 sera from 16 different wildlife species, most of them (92.6%) herbivores, were analysed by Enzyme-linked immunosorbent assay (ELISA) for IgG antibodies against *Brucella* spp, *C. burnetii* and RVFV. Further, 280 of these were tested by PCR to identify *Brucella* species.

Results

Of the 16 wildlife species tested, 15 (93.8%) were seropositive for at least one of the pathogens. Mean seropositivities were 18.9% (95% CI: 15.0–23.3) for RVFV, 13.7% (95% CI: 10.3–17.7) for *Brucella spp* and 9.1% (95% CI: 6.3–12.5) for *C. burnetii*. Buffaloes (n = 269) had higher seropositivity for *Brucella* spp. (17.1%, 95% CI: 13.0–21.7%) and RVFV (23.4%, 95% CI: 18.6–28.6%), while giraffes (n = 36) had the highest seropositivity for *C. burnetii* (44.4%, 95% CI: 27.9–61.9%). Importantly, 23 of the 93 (24.7%) animals positive for at least

design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

one pathogen were co-exposed, with 25.4% (18/71) of the positive buffaloes positive for brucellosis and RVFV. On molecular analysis, *Brucella* DNA was detected in 46 (19.5%, CI: 14.9–24.7) samples, with 4 (8.6%, 95% CI: 2.2–15.8) being identified as *B. melitensis*. The Fisher's Exact test indicated that seropositivity varied significantly within the different animal families, with *Brucella* (p = 0.013), *C. burnetii* (p = <0.001) and RVFV (p = 0.007). Location was also significantly associated (p = <0.001) with *Brucella spp.* and C. *burnetii* seropositivities.

Conclusion

Of ~20% of Kenyan wildlife that are seropositive for *Brucella* spp, *C. burnetii* and RVFV, almost 25% indicate co-infections with the three pathogens, particularly with *Brucella spp* and RVFV.

Author summary

Infection of an animal with more than one pathogen may result into a more devastating impact. Only few studies have investigated co-infection with multiple pathogens in wild-life, despite their key role as reservoirs of zoonotic diseases. Therefore, there is need for investigations on co-infection with neglected zoonotic pathogens in wildlife to inform prevention and control approaches and reduce disease impact in wildlife and potential transmission of these pathogens between wildlife, livestock, and humans. This study assessed co-exposure of various Kenyan wildlife species with three zoonotic pathogens, including *Brucella spp, Coxiella burnetii* and Rift Valley fever virus (RVFV). Results from this study revealed widespread, but varied exposure levels to the three individual pathogens within the several wildlife species. Likewise, the study also found the presence of co-exposure with the three pathogens. The findings from this study points to the need for establishment of surveillance and control programmes that target multiple pathogens in the wildlife populations to reduce the risk of transmission of infectious pathogens in wildlife and their zoonotic transmission.

Introduction

Zoonotic infections remain a key global threat, with emergence of new zoonoses such as Marburg and Ebola, as well as the resurgence and persistence of existing zoonotic infections, leading to devastating social, economic and health outcomes [1,2]. Human-animal interactions underpin the transmission of zoonoses, with the zoonotic pathogens being mobilized between vertebrate animals (both domestic and wildlife) and humans through various routes including direct contact with infected hosts, or indirect contact via the food chain, the environment or intermediate vectors such as ticks and mosquitoes [3]. Wildlife have been implicated in the emergence, maintenance and spillage of over 70% of zoonotic diseases [4–7]. Multisectoral, One Health strategies are thus recommended to predict, mitigate and control zoonotic infections [2].

Although many zoonotic diseases have a global distribution, several zoonoses categorized as "neglected zoonotic diseases" (NZDs), are associated with poverty, and huge public health and economic burden amongst the global poor [8]. Developing countries in Africa and Asia

have higher disease burden of these NZDs such as brucellosis, Q fever and Rift Valley fever, which are also classified as extremely dangerous pathogens (EDPs) [9–11]. The prevalence of these EDPs is usually underestimated and the risk levels remain understudied. Hence their control and prioritization in surveillance programmes in most developing countries remains underfunded [1,8]. Thus the first step towards understanding these zoonotic pathogens in a One Health framework is to understand their prevalence in wildlife populations.

Brucellosis, Q fever and RVF are associated with acute undifferentiated febrile illness in humans, which may progress to chronic disease with systemic manifestations [12–15]. In livestock, these three diseases cause abortion storms, stillbirths, premature and/or weak offspring, with significant reproductive and economic losses to farmers [13,16,17]. Humans acquire the causative pathogens through direct contact with tissues or body fluids of infected animals e.g. aborted materials or carcasses, aerosols in the case of Q fever and consumption of unpasteurized milk and dairy products in the case of brucellosis.

Increasing reports point to the role of wildlife as potential reservoirs for *Brucella* spp, *Coxiella burnetti* and RVF virus, although systematic studies on the ecology and epidemiology of these pathogens in wild animals remains limited. Recently, Simpson and co-workers [18], reported brucellosis in many species of wildlife in Africa, with buffalo, being implicated as maintenance hosts for *Brucella* species. Although *C. burnetii* is known to infect wildlife species [19], this pathogen was not detected in a previous survey that screened variety of wildlife species in Kenya [20]. In the case of RVF virus, little is known on the role of wildlife, although some studies suggest that wild animals may constitute a reservoir system for RVF virus during inter-epidemic seasons and in enzootic areas [21,22].

While several studies have shown exposure and potential circulation of Brucella spp, Coxiella burnetti and RVF virus in various wildlife species, the evidence for co-exposure of multiple species of these pathogens within wildlife populations, remains unexplored. Simultaneous spillover of multiple pathogens from wildlife to livestock and humans are likely to have implications on disease pathogenesis, disease course and health outcomes [23]. Co-infections may alter virulence of pathogens and subsequent disease outcomes in the hosts [23-25]. Across a wide range of pathogen taxa, co-infections generally lead to worse health outcomes for hosts and increase within host pathogen titers, altering transmission ecologies. Understanding these dynamics is therefore useful not only in furthering our knowledge of wildlife in the epidemiology of Q fever, RVF, and brucellosis, but also in calibrating existing risk models for these diseases in various contexts. Hence, the current study analyzed the seropositivity of Q fever, RVF, and brucellosis in 16 wildlife species belonging to the orders Artiodactyla, Perissodactyla and Carnivora from various parts of Kenya. We performed a retrospective study where samples collected during routine animal immobilizations for interventions due to traumatic injuries, translocation and collaring were tested. The findings give preliminary information on occurrence of co-infections in wildlife by geographic region, which has not been previously reported. We examined serological evidence of exposure to single and multiple pathogens, as well as molecular analysis of Brucella spp. in the different wildlife species.

Materials and methods

Study area and samples

A total of 363 archived sera samples from 16 wildlife species (Artiodactyla, Perissodactyla and Carnivora) were obtained from the Kenya Wildlife Service (KWS) in Nairobi. These samples were collected during routine veterinary interventions and disease surveillance activities conducted by the KWS between 2010 and 2021, and stored at -20°C. The samples originated from eight wildlife areas in Kenya including Central Rift area, Maasai Mara ecosystem, Coastal area,





https://doi.org/10.1371/journal.pntd.0010596.g001

Tsavo ecosystem, Laikipia-Samburu ecosystem, Amboseli ecosystem, Nairobi National park, and surrounding areas and Wajir area (Fig 1).

Serology

The serological testing of sera samples to detect the presence of antibodies against *Brucella*, *Coxiella burnetii* and RVFV was done using Enzyme-linked immunosorbent assays (ELISA), at the International Livestock Research Institute (ILRI), in Nairobi, Kenya. Individual serum samples were screened in duplicates for antibodies against *Coxiella burnetii*, RVFV and *Brucella*. The RVFV ELISA assay was done using ID screen Rift Valley Fever Competition Multispecies ELISA kit (IDvet innovative diagnostics, France) whereas ID screen Brucellosis Serum Indirect Multispecies ELISA kit (IDvet innovative diagnostics, France) was used for *Brucella*, and PrioCHECK Ruminant Q Fever AB Plate ELISA Kit (Applied Biosystems, Thermo Fisher Scientific) for *C. burnetii*. All the assays were done as per the manufacturer's instructions and the optical densities were read at wave lengths specified for each kit using BioTek ELISA reader (Synergy HT, BioTek Winooski, VT 05404 United States).

Molecular analysis using Real time PCR to detect Brucella species

Further molecular analysis were done to identify the circulating *Brucella* species among animals. Total DNA was extracted from 280 serum samples using QIAamp DNA Blood Mini Kit following the manufacturer's instructions except for elution which was done using 70µl of buffer AE. The purity and quantity of eluted DNA was determined using NanoDrop 1000 UV-Vis spectrophotometer (Thermo Scientific, Waltham, MA, USA) and stored at -20°C. *Brucella* genus was determined by Real-time PCR targeting *Bcsp31* using the primers and probes listed in <u>S1 Table</u>. The samples whose amplification threshold value was less than 40 were classified as PCR positive for *Brucella*. Multiplex Real time PCR was done on PCR-positive samples using species-specific primers and probes previously developed for *B. melitensis*, *B. abortus* and *B. suis* (<u>S1 Table</u>). Besides the test samples, each assay included a non-template (negative control), and a positive control DNA for each *Brucella* species.

The PCR amplifications were done using 2x Luna Universal Probe qPCR master mix (New England BioLabs, MA, USA), in a 20 μ l reaction volume as previously described [26]. Amplification was performed on QuantStudio 5 Real-time PCR System (Life Technologies Inc.) with the following conditions: Initial DNA denaturation at 95°C for 1 min, followed by 42 cycles of denaturation at 95°C for 15 secs and 1 min of annealing/amplification at 57°C.

Statistical analysis

Data were recorded and cleaned in Microsoft Excel version 2018 (IBM, California). All statistical analysis were performed in R statistical software version 3.6.3 [27]. The packages *DescTools* and *gmodels* within R, were used to perform all descriptive analysis including the estimation of positivity rates and 95% confidence intervals. The independent factors assessed for their association with the animal-level seropositivity status of the three-targeted pathogens included; age category, sex, location and animal species were determined using the Fisher's Exact test. However, some of the data had incomplete information on categorical factors; 71.1% of the records had information on animal species and location, while 58.1% and 48.8% had information on animal sex and age, respectively. The number of records varied by species. The varied numbers of the responses recorded for the different variables limited the use of multivariable analysis. Seropositivites were therefore calculated for only those species that had at least 30 samples, with exception for the animal categories. Tsavo ecosystem and Wajir area were excluded from the risk factor analysis due to low numbers of animals from these two regions.

Ethical statement

The samples were collected by KWS during their routine surveillance and animal translocation activities and as such no ethical approval was required for the work.

Results

Wildlife population summary and Elisa esults

A total of 363 sera samples, comprising 199 samples from buffaloes (*Syncerus caffer*), 36 giraffes (*Giraffa camelopardalis*), 21 zebras (*Equus burchelli*), 17 elands (*Taurotragus*), 15 oryxes (*Oryx beisa*), 11 waterbucks (*Kobus ellipsiprymnus*), 11 gazelles (*Gazella spp*), 9 impalas (*Aepyceros melampus*), 8 cheetahs (*Acinonyx jubatus*), 8 elephants (*Loxodonta africana*), 8 warthog (*Phacochoerus aethiopicus*), 7 rhinos (*Dicerosbicornis*), 5 lions (*Panthera leo*), 4 wildebeests (*Connochaetes taurinus*), 3 hartebeest (*Alcelaphus buselaphus coxii*), and 1 leopard (*Panthera pardus*) were analyzed. The samples were sourced from Maasai Mara ecosystem (n = 100), Coastal area (n = 54), Central Rift area (n = 48), Laikipia-Samburu ecosystem

(n = 32), Southern ecosystem (n = 10) and Wajir area (n = 3). The rest of the samples (n = 103) did not have information on their source. Of all the 211 samples with information on animal sex, 119 (56.4%) were males while 92 (43.6%) were females.

Amongst the three tested pathogens, RVFV had the highest seropositivity of 18.9% (95% CI: 15.0–23.3), followed by *Brucella* 13.7% (95% CI: 10.3–17.7) and *Coxiella burnetii* 9.1% (95% CI: 6.3–12.5) (Table 1). All except one (15/16) of the wildlife species had detectable antibodies against one or more of the three pathogens (Table 1), with only hartebeest (n = 3) testing negative for all the pathogens.

Factors associated with the seropositivity of Brucella

Positivity rates differed significantly between locations for *Brucella* (Fisher's Exact Test, P = < 0.001). A relatively higher proportion was recorded among animals sampled in Maasai Mara ecosystem 33.4% (33/97), Central Rift area 19.1% (9/47), and none from both Laikipia-Samburu ecosystem (0/29) and Coastal area (0/34) (Table 2 and Fig 1). Exposure levels for *Brucella* also differed significantly by animal families (Fisher's Exact Test, p = < 0.013), with antibodies against *Brucella* being detected more in *Bovidae* 17.1% (46/269) than Giraffidae 2.8% (1/36). No association was also observed between *Brucella* positivity with sex (Fisher's Exact Test, p = 0.864), and age category (Fisher's Exact Test, p = 0.571).

Factors associated with the seropositivity of C. burnetii

Positivity rates differed significantly between locations for *C. burnetii* (Fisher's Exact Test, p = <0.001). The Southern ecosystem had the highest positivity rate of 81.8% (9/11), Laikipia-

Table 1. Summary of the number of wildlife species tested and proportions of seropositive animals for Brucella, C. burnetii and RVFV.

Animals				Brucella		C. burnetii		RVFV	
Categories	Families	Species	Total no. Tested	No. positive	% positive (95% CI)	Number Positive	% positive (95% CI	Number of positive	% positive (95% CI)
Herbivores	Bovidae	Buffaloes	199	44	22.1 (16.5–28.)	5	2.5 (0.8-5.7)	41	20.6 (15.2–26.9)
		Eland	17	2		1		8	
		Wildebeest	4	0		0		1	
		Hartebeest	3	0		0		0	
		Gazelle	11	0		1		0	
		Impala	9	0		1		3	
		Waterbuck	11	0		3		0	
		Oryx	15	0		0		10	
		Total	269	46		11	4.1 (2.2–6.4)	63	23.4 (18.6–28.6)
	Giraffidae	Giraffe	36	1		16	44.4 (27.9–61.9)	2	5.6 (0.7-18.70
	Suidae	Warthog	8	1		0		0	
	Elephantidae	Elephant	8	0		0		2	
	Equidae	Zebra	21	0		3		0	
	Rhiocerotidae	Rhino	7	0		3		1	
	Total		349	48	13.7 (10.3–17.2)	33	9.4 (6.5–12.3)	68	19.4 (15.5–23.6)
Carnivores	Felidae	Leopard	1	1		0		0	
		Lion	5	1		0		0	
		Cheetah	8	0		0		1	
		Total	14	2	14.3 (7.1–35.2)*	0	0.0	1	7.1 (0.0–19.1)*
Overall 363			50	13.7 (10.3–17.7)	33	9.1 (6.3–12.5)	69	18.9 (15.0–23.3)	

Key:

* Low numbers of animals. Therefore, positivity estimates should be treated with caution

https://doi.org/10.1371/journal.pntd.0010596.t001

Factors		Brucella		C. burnetii		RVFV	
Variable	Category	Positivity	p-value	Positivity	p-value	positivity	p-value
Sex	Male	5% (5/100)	0.864	16% (16/100)	0.021	15% (15/1100)	0.792
	Female	7.9% (6/76)		5.3% (4/76)		18.4% (14/76)	
Families	Bovidae	17.1% (46/269)	0.013	4.1% (11/269)	< 0.001	23.4% (63/269)	0.007
	Giraffidae	2.8% (1/36)		44.4% (16/36)		5.6% (2/36)	
Age category	Sub-adult	0.0% (0/5)		0.0% (0/5)		0.0% (0/5)	
	Adult	5.7% (5/87)	0.751	14.9% (13/87)	0.458	8.0% (7/87)	0.667
Location	Southern ecosystem	0.0% (0/11)	< 0.001	81.8% (9/11)	< 0.001	0.0% (0/11)	0.130
	Central Rift area	19.1% (9/47)		6.4% (3/47)		29.8% (14/47)	
	Coastal area	0.0% (0/34)		5.9% (2/34)		38.2% (13/34)	
	Laikipia-Samburu ecosystem	0.0% (0/29)		10.3% (3/29)		27.6% (8/29)	
	Maasai Mara ecosystem	34.0% (33/97)		1.0% (1/97)		24.7% (24/97)	

Table 2. Factors associated with Brucella, C.burnetii and RVFV positivity.

https://doi.org/10.1371/journal.pntd.0010596.t002

Samburu ecosystem 10.3% (3/29), Central Rift 6.4% (3/47), Coastal area 5.9% (2/34), and Maasai Mara 1.0% (1/97). There was also significant differences in the positivity rates observed between the various animal families (Fisher's Exact Test, p <0.001). *Coxiella burnetii* antibodies were detected in *Giraffidae* 44.4% (16/36), and *Bovidae* 4.1% (11/269). Likewise, a higher proportion of positivity was observed in males 16% (16/100) compared to females 5.3% (4/76), (Fisher's Exact Test, p = 0.021). However,there was no association observed between the seropositivity of *C.burnetii* and age category (Fisher's exact test, p = 0.458).

Factors associated with seropositivity of RVF

The positivity status differed by animal families (Fisher's Exact Test, p = 0.007). *Bovidae* 23.4% (63/269), and *Giraffidae* 5.6% (2/36). No statistical association was found between RVF positivity with location (Fisher's Exact Test, p = 0.130), animal sex (Fisher's Exact Test, p = 0.792) and age category (Fisher's Exact Test, p = 0.667).

Brucella, C. burnetii and RVFV co-exposure in wildlife species

Evidence of co- exposure with antibodies against *Brucella*, *C. burnetii* and RVFV was observed in different wildlife species. Of 93 animals positive for at least one pathogen, 23 (24.7%) were co-exposed, with 25.4% (18/71) of the positive buffaloes positive for brucellosis and RVF. The overall co-exposure with antibodies from any of the three pathogens was 24.7% (95% CI: 16.4– 34.8). A relatively higher co-exposure was detected between *Brucella* and RVFV 22.7% (95% CI: 14.4–32.9), followed by *C. burnetii* and RVFV14.3% (95% CI: 1.7–42.8). Notably, there was no co-exposure between *Brucella* and *C. burnetii* antibodies for any species, whereas RVFV had co-exposure with either *Brucella* and *C. burnetii* (n = 1) (Table 3).

Molecular analysis for detection of Brucella spp

Brucella DNA was detected in eight out of all the 16 analyzed wildlife species. An aggregate of 46 samples were positive for *Brucella* species DNA by PCR (Table 4). A total of four samples (3 buffaloes and 1 giraffe) tested positive for *B. melitensis*, while none of the samples amplified with *B. suis* and *B. abortus* species targets (S1 Fig).

Discussion

This study moved from the widely-used approach of investigating single pathogens to a simultaneous investigation of multiple pathogens, each with a different transmission mode, to fill

Wildlife species		<i>C. burnetii</i> and RVF virus		<i>Brucella</i> RVF viru	and 15	C. burnetii and Brucella	Co-expos pathoger	Co-exposure with any of the three pathogens	
	Total N	N	% positive (95% CI)	Total n	% positive (95% CI)	Total n	Total n	% positive (95% CI)	
Buffaloes	71	0	0.0	18	25.4 (15.7–37.1)	0	18	25.3 (15.8-37.1)	
Eland	9	1	11.1(0.3-48.2)	1	11.1 (0.3-48.2)	0	2	22.2 (2.8-60.0)	
Giraffe	8	0	0.0	1	12.5 (0.3-52.6)	0	1	12.5 (0.3-52.6)	
Rhino	2	1	50 (1.3-98.7)	0	0.0	0	1	50.0 (1.3-98.7)	
Impala	3	1	33.3 (0.8–90.6)	0	0.0	0	1	33.3 (0.8–90.6)	
Total	93	3	14.3 (1.7-42.8)	20	22.7 (14.4-32.9)	0	23	24.7 (16.4-34.8)	

Table 3. Showing proportion of co-exposure with *Brucella*, *C. burnetii* and RVFV antibodies in different wildlife species. Only wildlife species that had an exposure with at least one of the pathogens were included in the analysis for co-exposure.

Key: N total number of animals that tested positive for any of the three pathogens (both single and co- exposure), n number of co-exposure cases, CI Confidence Interval.

https://doi.org/10.1371/journal.pntd.0010596.t003

the existing knowledge gap on the co-exposure to one viral (RVFV) and two bacterial (*Brucella and C. burnetii*) pathogens in wildlife populations in Kenya. Co-infection is widely recognized as one of the leading drivers of pathogen evolution and epidemiology [23,28], with more devastating epidemics observed in co-infected populations. We present evidence of co-exposure to *Brucella, C. burnetii* and RVFV exposure in various wildlife species in Kenya. We also observed variations in positivity rates of each of the three pathogens as a single exposure. However, co-occurrence between *Brucella* and *C. burnetii* antibodies were not detected in our data. We also found evidence of *Brucella* DNA in several wildlife species and *B. melitensis* DNA in buffaloes and giraffes.

We found an overall co-occurrence of *Brucella*, *C. burnetii* and RVFV antibodies in wildlife to be 24.7%, which is higher than the recently reported 15.0% of co-occurrence of the same pathogens in camels in Kenya [29]. Interestingly, we observed co-exposures with RVFV as common, more frequently with *Brucella* and less frequently with *C. burnetii*. These findings suggests that wildlife species could play a significant role in maintaining and transmitting multiple zoonotic pathogens at the same time [4,7]. Indeed wildlife species are known to be reservoirs of many zoonotic pathogens, which may spillover to either livestock and humans directly, or via an appropriate intermediate host [30]. Further, the increasing human population, coupled with shrinking and degraded wildlife habitats, are increasing wildlife population

Table 4. Showing wildlife species with the genus Brucella PCR positive results. The total and distribution of samples positive for genus Brucella, B. abortus, B. melitensis and B. suis.

Wildlife species		Total number of positi	Total number of positive samples						
Species	Total tested	Genus Brucella.	Genus Brucella. B. abortus		B. melitensis B. suis				
Buffaloes	177	32	0	3	0				
Cheetah	6	1	0	0	0				
Eland	9	3	0	0	0				
Elephant	5	2	0	0	0				
Giraffe	20	5	0	1	0				
Lion	4	1	0	0	0				
Oryx	6	1	0	0	0				
Warthog	8	1	0	0	0				
Total	235	46 (19.5%)	0 (0.0%)	4 (8.6%)	0 (0.0%)				

https://doi.org/10.1371/journal.pntd.0010596.t004

densities and could be responsible for transmitting infectious diseases and the relatively higher rate of co-occurrence of pathogens [31].

Our study found no evidence of co-exposure to *Brucella* and *C. burnetii*, despite their single occurrence in the same wildlife populations. Nevertheless, this coinfection was reported in domestic animals in Guinea [32] and therefore, we cannot entirely rule out the possibility of co-exposure to the two pathogens in wild species. Further, our ongoing unpublished studies have also identified *Brucella* and *C. burnetii* co-exposure in livestock belonging to the *Bovidae* family, to which some wildlife species investigated here belong. Nevertheless, these findings highlight the need to fully investigate the potential microbial interactions amongst *Brucella* spp. and *C. burnetii* in wildlife hosts.

We observed a higher rate of seropositivity to *Brucella* species (33%) and RVFV (20.4%) in the Maasai Mara ecosystem relatively to the rest of the areas in this study, suggesting an ecology permissive for transmission of both pathogens. Brucella species can survive in the environment for several months in cool and moist conditions which are also necessary for cryptic RVFV vectors breeding [33]. In addition, scavenging of contaminated placentas and weak offspring by predators, could be an additional risk factor for environmental transmission through mechanical dispersal for this pathogen. Other studies have shown the positivity rate of Brucella was similar for cattle (31.8%) and human (40.8%) in the same region, where there are intense wildlife and livestock interaction [26,34]. This highlights the interaction between wildlife, livestock, and humans in Brucella disease ecology. High wildlife and livestock population densities, climatic and land-use changes, and the perennial human-wildlife conflict in this ecosystem all lead to increasing wildlife-livestock-human interaction, making it a disease transmission hotspot posing public health risks [35]. This and other similar ecosystems could be ideal for investigating risk factors associated with zoonosis co-occurrence/co-infection dynamics. Although close contact between wildlife, human and livestock enable transmission of zoonotic pathogens [34], the directionality of transmission, could not be deduced from the current study; are these wildlife acting as reservoir hosts, or are they transmission dead ends. Therefore, further research involving a multidisciplinary, One Health approach is needed to fully explore the complex transmission dynamics of zoonotic pathogens that we studied and others in human-wildlife-livestock interfaces to inform integrated program for concurrent control of zoonoses.

The exposure levels for both *Brucella spp* and *C. burnetii* differed significantly, depending on the study locations, corroborating with reports of clustering of *Brucella spp*, *C. burnetii* and cases of other infectious diseases in domesticated animals [26,34]. However, a statistical association between location and animal species with exposure levels to RVFV antibodies was not found, contrary to previous reports [36,37]. Thus, the observed differences in exposure levels in different locations perhaps arose from spatial differences in environmental factors and the animal species present in various locations, considering that positivity rates also differed significantly in the different animal species. Future studies should target specific host species with unbiased sampling to better identify driving factors of co-exposure or co-infection. Regardless of the limitations of using archived samples in the current study, these findings have implications on the spread of zoonotic disease through wildlife movement, e.g., through natural migrations and wildlife translocations. Policies need to integrate disease risk analyses during wildlife translocations to prevent the introduction of pathogens to new areas or to spread to other wildlife, humans, and livestock in the destinations.

By highlighting evidence of exposure to three zoonotic pathogens, this study draws attention to potential anthropogenic activities that may result in spillovers of brucellosis, RVF, and Q fever infections from wildlife to humans and livestock. Emergence and spread of zoonotic infections is driven by several anthropogenic factors, including habitat fragmentation and degradation, increasing consumption of animal-derived foods; agricultural intensification; human encroachment in wildlife habitats and climate change, which cumulatively increase human-animal interactions. The hunting and trade of wildlife for meat may pose a heightened threat for potential exposure of humans to *Brucella*, *C. burnetii* and *RVFV*, for which slaughter, handling, and consumption of contaminated carcasses are major risk factors [35,38–39]. Our findings that wild ungulates (especially buffalo) had the highest exposure rates for all the pathogens, coupled with the fact that wild ungulate species are predominantly hunted for bushmeat in eastern Africa [36,40–43], underpin the potential risk of these and other anthropogenic activities in wildlife zoonotic transmission. Additional studies geared specifically to investigate these pathogens from a One Health perspective are warranted. These will target the interfaces between wildlife, livestock, and humans for instance, testing for infection with multiple pathogens where water and land resources are shared.

The results from molecular analysis of *Brucella* species demonstrated the presence of *Brucella* DNA in different wildlife hosts and further confirmed possible circulation of the zoonotic species of *Brucella* (*B. melitensis*) in buffaloes and giraffe. However, there were disparities between the serological and molecular findings concerning brucellosis in that not all animals positive for Brucella antibodies were positive by DNA analysis. These findings augment Alsubaie and co-workers' [44] findings where serological diagnosis in human patients did not correlate with pathogen isolation by culture. This could be partly due to transient presence of *Brucella* antigens in blood where they are detected only for a short period around abortions, whereas antibodies persist for longer durations in the absence of antigens [45]. Caution should therefore be exercised in the choice of the diagnostic assay, based on the study objective(s). Nonetheless, the molecular findings of *Brucella melitensis* DNA in buffalo and giraffe further strengthen the evidence that wildlife harbors the pathogens and may have active infections that could be transmissible to susceptible hosts, especially humans, given that *B. melitensis* is a major cause of bloodstream infections and febrile illness in pastoral communities in eastern Africa [26,46].

This study utilized archived samples collected retrospectively over 10 years, which limited our ability to fully evaluate the contribution of different variables to the epidemiology of zoo-notic diseases under investigation. Therefore, active nationwide surveillance with and systematic sampling approach could be conducted to fully explore the epidemiology of *Brucella*, *Coxiella* and RVFV in wildlife populations in Kenya, preferably augmented with laboratory surveillance.

Supporting information

S1 Table. Oligonucleotide primers and probes. (TIF)

S1 Fig. Amplification plots for *B. abortus* and *B. melitensis.* (TIF)

Acknowledgments

We acknowledge the contribution of the Kenya Wildlife Service field and laboratory staff for supporting data and sample processing. We also appreciate Prof. Heinrich Neubauer and Dr. Falk Melzer from Friedrich-Loeffler-Institute, for offering the controls for this study.

Author Contributions

Conceptualization: Francis Gakuya, Kariuki Njenga, Bernard Bett.

Data curation: James Akoko.

Formal analysis: James Akoko, Collins Ouma, Daniel Nthiwa, Bernard Bett.

Funding acquisition: Bernard Bett.

Investigation: James Akoko, Lillian Wambua, Richard Nyamota, Bernard Ronoh, Isaac Lekolool, Athman Mwatondo, Mathew Muturi, Daniel Nthiwa, John Gachohi.

Methodology: James Akoko, Lillian Wambua, Richard Nyamota.

Project administration: Francis Gakuya.

Supervision: Francis Gakuya.

Validation: James Akoko.

Visualization: James Akoko.

- Writing original draft: James Akoko, Lillian Wambua, Richard Nyamota, Earl Middlebrook, Jeanne Fair, John Gachohi.
- Writing review & editing: Francis Gakuya, James Akoko, Lillian Wambua, Richard Nyamota, Bernard Ronoh, Isaac Lekolool, Athman Mwatondo, Mathew Muturi, Collins Ouma, Daniel Nthiwa, Earl Middlebrook, Jeanne Fair, John Gachohi, Kariuki Njenga, Bernard Bett.

References

- Rahman T, Sobur A, Islam S, Ievy S, Hossain J, El Zowalaty ME, et al. Zoonotic Diseases: Etiology, Impact, and Control. Microorganisms. 2020 Sep; 8(9):1405. Available from: <u>https://www.mdpi.com/</u> 2076-2607/8/9/1405.
- 2. FAO, OIE and WHO. Taking a Multisectoral, One Health Approach: A Tripartite Guide to Addressing Zoonotic Diseases in Countries. 2019.
- Christou L. The global burden of bacterial and viral zoonotic infections. Clinical Microbiology and Infection. 2011 Mar; 17(3):326–30. Available from: https://linkinghub.elsevier.com/retrieve/pii/ S1198743X14638635. https://doi.org/10.1111/j.1469-0691.2010.03441.x PMID: 21129102
- Kruse H, Kirkemo AM, Handeland K. Wildlife as Source of Zoonotic Infections. Emerg Infect Dis. 2004 Dec; 10(12):2067–72. Available from: http://wwwnc.cdc.gov/eid/article/10/12/04-0707_article.htm. https://doi.org/10.3201/eid1012.040707 PMID: 15663840
- Allen T, Murray KA, Zambrana-Torrelio C, Morse SS, Rondinini C, Di Marco M, et al. Global hotspots and correlates of emerging zoonotic diseases. Nat Commun. 2017 Dec; 8(1):1124. Available from: http://www.nature.com/articles/s41467-017-00923-8. https://doi.org/10.1038/s41467-017-00923-8 PMID: 29066781
- Jones KE, Patel NG, Levy MA, Storeygard A, Balk D, Gittleman JL, et al. Global trends in emerging infectious diseases. Nature. 2008 Feb; 451(7181):990–3. Available from: http://www.nature.com/ articles/nature06536. https://doi.org/10.1038/nature06536 PMID: 18288193
- Rhyan JC, Spraker TR. Emergence of Diseases From Wildlife Reservoirs. Vet Pathol. 2010 Jan; 47 (1):34–9. Available from: <u>http://journals.sagepub.com/doi/10.1177/0300985809354466</u>. PMID: 20080482
- Grace D, Mutua F, Ochungo P, Kruska RL, Jones K, Brierley L, et al. Mapping of poverty and likely zoonoses hotspots. 2012.
- Mableson HE, Okello A, Picozzi K, Welburn SC. Neglected Zoonotic Diseases—The Long and Winding Road to Advocacy. Njenga MK, editor. PLoS Negl Trop Dis. 2014 Jun; 8(6):e2800. Available from: https://dx.plos.org/10.1371/journal.pntd.0002800. https://doi.org/10.1371/journal.pntd.0002800 PMID: 24901769
- Njeru J, Henning K, Pletz MW, Heller R, Neubauer H. Q fever is an old and neglected zoonotic disease in Kenya: a systematic review. BMC Public Health. 2016 Dec; 16(1):297. Available from: http://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-016-2929-9. http://bmcpublichealth.biomedcentral.com/articles/10.1186/s12889-016-2929-9. https://doi.org/10.1186/s12889-016-2929-9. <a href="https://doi.org/1

- Dean AS, Crump L, Greter H, Schelling E, Zinsstag J. Global Burden of Human Brucellosis: A Systematic Review of Disease Frequency. Carabin H, editor. PLoS Negl Trop Dis. 2012 Oct; 6(10):e1865. Available from: https://dx.plos.org/10.1371/journal.pntd.0001865. https://doi.org/10.1371/journal.pntd. 0001865 PMID: 23145195
- Dean AS, Crump L, Greter H, Hattendorf J, Schelling E, Zinsstag J. Clinical Manifestations of Human Brucellosis: A Systematic Review and Meta-Analysis. Carabin H, editor. PLoS Negl Trop Dis. 2012 Dec 6; 6(12):e1929. Available from: https://dx.plos.org/10.1371/journal.pntd.0001929. https://doi.org/10. 1371/journal.pntd.0001929 PMID: 23236528
- Vanderburg S, Rubach MP, Halliday JEB, Cleaveland S, Reddy EA, Crump JA. Epidemiology of Coxiella burnetii Infection in Africa: A OneHealth Systematic Review. Njenga MK, editor. PLoS Negl Trop Dis. 2014 Apr; 8(4):e2787. Available from: https://dx.plos.org/10.1371/journal.pntd.0002787. https:// doi.org/10.1371/journal.pntd.0002787 PMID: 24722554
- Clark MHA, Warimwe GM, Di Nardo A, Lyons NA, Gubbins S. Systematic literature review of Rift Valley fever virus seroprevalence in livestock, wildlife and humans in Africa from 1968 to 2016. Barker CM, editor. PLoS Negl Trop Dis. 2018 Jul; 12(7):e0006627. Available from: https://dx.plos.org/10.1371/journal. pntd.0006627. https://doi.org/10.1371/journal.pntd.0006627 PMID: 30036382
- LaBeaud AD, Pfeil S, Muiruri S, Dahir S, Sutherland LJ, Traylor Z, et al. Factors Associated with Severe Human Rift Valley Fever in Sangailu, Garissa County, Kenya. Williams M, editor. PLoS Negl Trop Dis. 2015 Mar; 9(3):e0003548. Available from: https://dx.plos.org/10.1371/journal.pntd.0003548. https://doi. org/10.1371/journal.pntd.0003548 PMID: 25764399
- 16. Mcdermott JJ, Grace D, Zinsstag J. Economics of brucellosis impact and control in low-income countries. Rev Sci Tech OIE. 2013 Apr; 32(1):249–61. Available from: https://doc.oie.int/dyn/portal/index.seam?page=alo&aloId=31526. https://doi.org/10.20506/rst.32.1.2197 PMID: 23837382
- Salifu SP, Bukari ARA, Frangoulidis D, Wheelhouse N. Current perspectives on the transmission of Q fever: Highlighting the need for a systematic molecular approach for a neglected disease in Africa. Acta Tropica. 2019 May; 193:99–105. Available from: https://linkinghub.elsevier.com/retrieve/pii/ S0001706X19300889. https://doi.org/10.1016/j.actatropica.2019.02.032 PMID: 30831112
- Simpson G, Thompson PN, Saegerman C, Marcotty T, Letesson JJ, de Bolle X, et al. Brucellosis in wildlife in Africa: a systematic review and meta-analysis. Sci Rep. 2021 Dec; 11(1):5960. Available from: http://www.nature.com/articles/s41598-021-85441-w. https://doi.org/10.1038/s41598-021-85441-w PMID: 33727580
- 19. González-Barrio D, Ruiz-Fons F. *Coxiella burnetii* in wild mammals: A systematic review. Transbound Emerg Dis. 2019 Mar; 66(2):662–71. Available from: <u>https://onlinelibrary.wiley.com/doi/abs/10.1111/tbed.13085</u>. PMID: 30506629
- Ndeereh D, Muchemi G, Thaiyah A, Otiende M, Angelone-Alasaad S, Jowers MJ. Molecular survey of Coxiella burnetii in wildlife and ticks at wildlife–livestock interfaces in Kenya. Experimental and Applied Acarology. 2017 Jul; 72(3):277–89. Available from: http://www.ncbi.nlm.nih.gov/pubmed/28593481 http://link.springer.com/10.1007/s10493-017-0146-6. https://doi.org/10.1007/s10493-017-0146-6 PMID: 28593481
- Britch SC, Binepal YS, Ruder MG, Kariithi HM, Linthicum KJ, Anyamba A, et al. Rift Valley Fever Risk Map Model and Seroprevalence in Selected Wild Ungulates and Camels from Kenya. Ikegami T, editor. PLoS ONE. 2013 Jun 28; 8(6):e66626. Available from: https://dx.plos.org/10.1371/journal.pone. 0066626. https://doi.org/10.1371/journal.pone.0066626 PMID: 23840512
- Rostal MK, Liang JE, Zimmermann D, Bengis R, Paweska J, Karesh WB. Rift Valley Fever: Does Wildlife Play a Role? ILAR Journal. 2017 Dec; 58(3):359–70. Available from: https://academic.oup.com/ ilarjournal/article/58/3/359/4100591. https://doi.org/10.1093/ilar/ilx023 PMID: 28985319
- Seabloom EW, Borer ET, Gross K, Kendig AE, Lacroix C, Mitchell CE, et al. The community ecology of pathogens: coinfection, coexistence and community composition. Ostfeld R, editor. Ecol Lett. 2015 Apr; 18(4):401–15. Available from: https://onlinelibrary.wiley.com/doi/10.1111/ele.12418. PMID: 25728488
- Clay PA, Rudolf VHW. How parasite interaction strategies alter virulence evolution in multi-parasite communities. Evolution. 2019 Nov; 73(11):2189–203. Available from: https://onlinelibrary.wiley.com/ doi/10.1111/evo.13843. PMID: 31506940
- Alizon S, van Baalen M. Multiple Infections, Immune Dynamics, and the Evolution of Virulence. The American Naturalist. 2008 Oct; 172(4):E150–68. Available from: <u>https://www.journals.uchicago.edu/</u> doi/10.1086/590958. PMID: 18702601
- Akoko JM, Pelle R, Lukambagire AS, Machuka EM, Nthiwa D, Mathew C, et al. Molecular epidemiology of Brucella species in mixed livestock-human ecosystems in Kenya. Sci Rep. 2021 Dec; 11(1):8881. Available from: http://www.nature.com/articles/s41598-021-88327-z. https://doi.org/10.1038/s41598-021-88327-z PMID: 33893352
- 27. R Core Team. An Introduction to R. Indusa and Commercial Training. 2008;10(1):11-8.

- Susi H, Barrès B, Vale PF, Laine AL. Co-infection alters population dynamics of infectious disease. Nat Commun. 2015 May; 6(1):5975. Available from: http://www.nature.com/articles/ncomms6975. https:// doi.org/10.1038/ncomms6975 PMID: 25569306
- 29. Muturi M, Akoko J, Nthiwa D, Chege B, Nyamota R, Mutiiria M, et al. Serological evidence of single and mixed infections of Rift Valley fever virus, Brucella spp. and Coxiella burnetii in dromedary camels in Kenya. Lin T, editor. PLoS Negl Trop Dis. 2021 Mar; 15(3):e0009275. Available from: https://dx.plos.org/10.1371/journal.pntd.0009275. https://doi.org/10.1371/journal.pntd.0009275 PMID: 33770095
- Reusken C, van der Plaats R, Opsteegh M, de Bruin A, Swart A. Coxiella burnetii (Q fever) in Rattus norvegicus and Rattus rattus at livestock farms and urban locations in the Netherlands; could Rattus spp. represent reservoirs for (re)introduction? Preventive Veterinary Medicine. 2011 Aug; 101(1– 2):124–30. Available from: https://linkinghub.elsevier.com/retrieve/pii/S0167587711001516. https://doi. org/10.1016/j.prevetmed.2011.05.003 PMID: 21640416
- Daszak P, Cunningham AA, Hyatt AD. Emerging infectious diseases of wildlife—Threats to biodiversity and human health. Science. 2000; 287(5452):443–9. <u>https://doi.org/10.1126/science.287.5452.443</u> PMID: 10642539
- 32. Troupin C, Ellis I, Doukouré B, Camara A, Keita M, Kagbadouno M, et al. Seroprevalence of brucellosis, Q fever and Rift Valley fever in domestic ruminants in Guinea in 2017–2019. BMC Vet Res. 2022 Dec; 18(1):64. Available from: https://bmcvetres.biomedcentral.com/articles/10.1186/s12917-022-03159-x. https://doi.org/10.1186/s12917-022-03159-x PMID: 35120506
- Kaden R, Ferrari S, Jinnerot T, Lindberg M, Wahab T, Lavander M. Brucella abortus: determination of survival times and evaluation of methods for detection in several matrices. BMC Infect Dis. 2018 Dec; 18(1):259. Available from: https://bmcinfectdis.biomedcentral.com/articles/10.1186/s12879-018-3134-5. https://doi.org/10.1186/s12879-018-3134-5 PMID: 29871600
- Nthiwa D, Alonso S, Odongo D, Kenya E, Bett B. Zoonotic Pathogen Seroprevalence in Cattle in a Wildlife–Livestock Interface, Kenya. EcoHealth. 2019 Dec; 16(4):712–25. Available from: http://link. springer.com/10.1007/s10393-019-01453-z. https://doi.org/10.1007/s10393-019-01453-z PMID: 31728795
- Alexander KA, Blackburn JK, Vandewalle ME, Pesapane R, Baipoledi EK, Elzer PH. Buffalo, Bush Meat, and the Zoonotic Threat of Brucellosis in Botswana. Schneider BS, editor. PLoS ONE. 2012 Mar 8; 7(3):e32842. Available from: https://dx.plos.org/10.1371/journal.pone.0032842. https://doi.org/10. 1371/journal.pone.0032842 PMID: 22412932
- **36.** Lwande OW, Paul GO, Chiyo PI, Ng'ang'a E, Otieno V, Obanda V, et al. Spatio-temporal variation in prevalence of Rift Valley fever: a post-epidemic serum survey in cattle and wildlife in Kenya. Infection Ecology & Epidemiology. 2015; 5(1):30106. https://doi.org/10.3402/iee.v5.30106 PMID: 26679561
- Evans A, Gakuya F, Paweska JT, Rostal M, Akoolo L, Van Vuren PJ, et al. Prevalence of antibodies against Rift Valley fever virus in Kenyan wildlife. Epidemiology and Infection. 2008; 136(9):1261–9. https://doi.org/10.1017/S0950268807009806 PMID: 17988425
- Katani R, Schilling MA, Lyimo B, Tonui T, Cattadori IM, Eblate E, et al. Microbial Diversity in Bushmeat Samples Recovered from the Serengeti Ecosystem in Tanzania. Sci Rep. 2019 Dec; 9(1):18086. Available from: http://www.nature.com/articles/s41598-019-53969-7. https://doi.org/10.1038/s41598-019-53969-7 PMID: 31792246
- 39. Ndengu M, Matope G, Tivapasi M, Pfukenyi DM, Cetre-Sossah C, De Garine-Wichatitsky M. Seroprevalence and associated risk factors of Rift Valley fever in cattle and selected wildlife species at the livestock/wildlife interface areas of Gonarezhou National Park, Zimbabwe. Onderstepoort j vet res. 2020 Apr 8; 87(1). Available from: https://ojvr.org/index.php/ojvr/article/view/1731. https://doi.org/10.4102/ ojvr.v87i1.1731 PMID: 32370521
- **40.** Ogutu JO, Owen-Smith N, Piepho HP, Said MY. Continuing wildlife population declines and range contraction in the Mara region of Kenya during 1977–2009. Journal of Zoology. 2011 Oct; 285(2):99–109. Available from: http://doi.wiley.com/10.1111/j.1469-7998.2011.00818.x.
- Ouso DO, Otiende MY, Jeneby M, Oundo JW, Bargul JL, Miller S, et al. Three-gene PCR and high-resolution melting analysis for differentiating vertebrate species mitochondrial DNA for forensic and biodiversity research pipelines. bioRxiv. 2019 May; Available from: <u>http://biorxiv.org/lookup/doi/10.1101/</u> 636605.
- 42. Rentsch D. The Nature of Bushmeat Hunting in the Serengeti Ecosystem, Tanzania: Socio-economic Drivers of Consumption of Migratory Wildlife. PhD thesis. 2011;(May):104.
- 43. Green DS, Zipkin EF, Incorvaia DC, Holekamp KE. Long-term ecological changes influence herbivore diversity and abundance inside a protected area in the Mara-Serengeti ecosystem. Global Ecology and Conservation. 2019; 20:e00697–e00697. Available from: https://doi.org/10.1016/j.gecco.2019.e00697.
- 44. Alsubaie SA, Turkistani SA, Zeaiter AA, Thabit AK. Lack of correlation of Brucella antibody titers with clinical outcomes and culture positivity of brucellosis. Trop Dis Travel Med Vaccines. 2021 Dec; 7(1):5.

Available from: https://tdtmvjournal.biomedcentral.com/articles/10.1186/s40794-021-00130-w. https:// doi.org/10.1186/s40794-021-00130-w PMID: 33531081

- 45. Ducrotoy MJ, Muñoz PM, Conde-Álvarez R, Blasco JM, Moriyón I. A systematic review of current immunological tests for the diagnosis of cattle brucellosis. Preventive Veterinary Medicine. 2018 Mar; 151:57–72. Available from: https://linkinghub.elsevier.com/retrieve/pii/S0167587717301484. https:// doi.org/10.1016/j.prevetmed.2018.01.005 PMID: 29496108
- 46. Bodenham RF, Lukambagire AS, Ashford RT, Buza JJ, Cash-Goldwasser S, Crump JA, et al. Prevalence and speciation of brucellosis in febrile patients from a pastoralist community of Tanzania. Sci Rep. 2020 Dec; 10(1):7081. Available from: http://www.nature.com/articles/s41598-020-62849-4. https://doi.org/10.1038/s41598-020-62849-4 PMID: 32341414